Author's response to reviews

Title: Intestinal ischemia following laparoscopic surgery

Authors:

Waleed Al-Khyatt (mzxwa@nottingham.ac.uk)
James Thomas (driithomas@gmail.com)
David J Humes (david.humes@nottingham.ac.uk)
Dileep N Lobo (dileep.lobo@nottingham.ac.uk)

Version: 5 Date: 10 September 2012

Author's response to reviews:

Dear Editor,

We have considered the suggestions regarding the paragraph about intra-abdominal hypertension as the etiology for intestinal ischemia the current knowledge about acute intestinal distress syndrome (AIDS). The necessary update has been performed and highlighted in red. The appropriate references have been added too.

Regards,

Waleed Al-Khyatt